Author's response to reviews

Title: The Effect of Adjuctive Chlorhexidine Mouthrinse on GCF MMP-8 And TIMP-1 Levels in Gingivitis: A Randomized Placebo-Controlled Study

Authors:

Oya Turkoglu (oya_t@yahoo.com)
Sema Becerik (semacinar@yahoo.com)
Taina Tervahartiala (taina.tervahartiala@helsinki.fi)
Timo Sorsa (timo.sorsa@helsinki.fi)
Gul Atilla (gulatilla@yahoo.com)
Gulnur Emingil (gemingil@yahoo.com)

Version: 4 Date: 2 May 2014

Author's response to reviews:

2 May 2014

Re: “The Effect of Adjuctive Chlorhexidine Mouthrinse on GCF MMP-8 and TIMP-1 Levels in Gingivitis: A Randomized Placebo-Controlled Study”

Dear editor,

In reference to your letter dated 16 April 2014 about the reviews of our manuscript, I am sending you our revised manuscript. I have underlined the changes I performed in the manuscript. We have submitted the amendments to editor at the following page.

We would very much appreciate the interest you have shown for our manuscript.

Sincerely yours,
Oya Türkoğlu, D.D.S., PhD.
Ege University, School of Dentistry,
Department of Periodontology,
35100 Izmir/Turkey
Fax: +90 232 3880325
e-mail: oya_t@yahoo.com
Mail address: Ege University, School of Dentistry,
Department of Periodontology,
Bornova, 35100 Izmir
Turkey
Point-to point Amendments to Editor
Please note that some copy-editing for language is required in this section.

As suggested by editor, we re-wrote the section in question as followings (please see page 18).

“OT and SB designed the study, collected the samples, commented on the data and wrote the manuscript. TT carried out the ELISA tests and commented on the data. TS, GA and GE came up with hypothesis, commented on the data and helped to write the manuscript.”